CENWP-OP-B                                                                                                                                     17 March 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 11 of 2003, which covers the period 09 through 15 March 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 129.4 to 169.2 kcfs.  Daily project tailwater elevation averaged 12.8’ to 16.2’ msl.  Daily project forebay elevation averaged 72.1’ to 74.8’ msl.  River temperature averaged 41 to 43° F.  Daily turbidity measured 3.0’ to 5.6’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are open .5’ during daylight hours in accordance to the FPP.  We spilled water for Spring Creek hatchery release from 1800 on 10 March until 0600 on 12 March.  Spill averaged 1.6 to 52.0 kcfs.

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  During fish passage season, the units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  Fish Unit 1 remains out of service for overhaul and stator rewind.

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  The following AWS valves were closed for cleaning this reporting week.

FV3-9 closed from 1248 until 1314 on 10 March and from 1455 until 1500 on 13 March.

2.2 STS/VBS Inspections.  No STS/VBSs inspections this reporting week. 

2.3 Dewatering and Fish Salvage.  No dewater or salvage activities this reporting week.

3.  RESEARCH.

3.1 University of Idaho/WDFW jaw tagging.  Steelhead and spring chinook will be tagged for studies involving the tooth tangle net fishery on the Columbia.

3.2 NMFS Lamprey.  Tracking continues.

3.3 PSMFC.  Smolt monitoring started 24 hour sampling at the JMF on 10 March.  One dead kelt, 44 steelhead, six salmon and one smolt slid off the end of the separator bars. 

3.4 USGS Juvenile Radiotracking.  USGS began installing equipment around the project.

3.5 WDFW Fish Counting.  16-hr video counts at Washington Shore and Bradford Island.

3.6 WDFW steelhead.  Trapping and tagging of winter steelhead continues.

4.  FISHWAYS 
4.1 Project biologists inspected 09-15 March this week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	See misc. 5.1
	

	3/9
	DSM2
	ERG open
	Debris plug in SMF bypass flume.
	See misc. 5.4
	

	3/9, 3/15
	Weir 37, 38
	Reading < 1.0’
	Unknown
	Operators to adjust
	

	3/10-12
	DSM2
	Low channel elevation
	Low forebay
	All orifices open, add in full open.
	


4.3 Fish Ladder Temperatures.  Temperatures are not recorded in the winter.

4.4 Sensor Calibration.  Calibration checked on 13 March.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Divers discovered the PH2 velocity meter was missing from the collection channel.  A new meter will be installed during the mid season dives this summer.

5.2 Cascade Island and B-Branch controls are in manual mode awaiting tailwater sensor calibration.  Electricians are awaiting software.   

5.3 On 24 January, the Project tailwater hard constraint decreased to 11.0’ msl for all hours.  The soft constraint went to 11.2’ msl between 0700-1900 hours.  This is for the protection of the chum redds around Ives Island.  On 17 February, BPA agreed to a hard tailwater constraint of 11.3’ msl and a soft constraint of 11.5’ msl when inflows to Bonneville are >120Kcfs.

5.4 At 2240 on 02 March, the PH1 orifice and sluice gates were closed.  Bulkheads will be installed this spring.

5.5 On 02 March the bypass flume was drained  to facilitate removing a debris plug comprised of woody debris and aquatic macrophytes.  The plug formed just above the lower switchgate.  A similar plug formed the morning of 05 March and the ERG was again opened and the flume drained.   Trash raking had been ongoing for a week when the plugs formed.  Biologists requested the ERG be left open until 10 March, when sampling of Spring Creek fish will begin.

5.6 At 1100 on 09 March, the new navlock went out of service for maintenance.   

5.7 At 0450 on 10 March, the ERG was closed and at 0650 the monitoring facility was put in sample mode.

5.8 Spring Creek released 7.6 million fall chinook on 08 March.  A teletype requested 50kcfs of spill for 36 hours from 1800 on 10 March until 0600 on 12 March.  Spill Bays 14, 6, and 3 were not functioning properly.  Adjacent bays were opened for Spring Creek spill.

5.9 The south monolith was put back in service on 10 March.  

5.10 The Project is currently operating under new guidelines for one fish unit operation.  Even with the mods to the floating orifice gates, we still have some trouble maintaining differential at the south end.  We will manipulate entrance gates to try to achieve proper differential.

5.11 Due to the low forebay restriction for navlock maintenance, DSM2 had low water elevations early in the week.  All available orifices were opened and the add-in opened 100%.  The restriction was lifted at 1143 on 12 March.  The DSM is back in criteria.

5.12 The main dam was on seal due to dive operations off Bradford Island on 11, 13 and 14 March.  Sediment samples were taken as part of the Bradford Island cleanup.

5.13 Contractor repairs to the AFF were completed on 14 March.

5.14 Debarking activities in PH2 gatewells have been ongoing all week.

5.15 DSM1 is running south as we are waiting for NMFS to complete repairs to their PIT box.  We expect to return the DSM to normal flow on Monday, 17 March.  

 JAMES R. MAHAR

                                 

                                                           Operations Manager
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